Author’s response to reviews

Title: The share of ultra-processed foods and the overall nutritional quality of diets in the US: evidence from a nationally representative cross-sectional study

Authors:

Euridice Martinez Steele (emar_steele@hotmail.com)

Barry Popkin (popkin@unc.edu)

Boyd Swinburn (boyd.swinburn@auckland.ac.nz)

Carlos Monteiro (carlosam@usp.br)

Version: 1 Date: 14 Nov 2016

Author’s response to reviews:

RTR